Author's response to reviews

Title: A model-based economic analysis of pre-pandemic vaccination cost-effectiveness

Authors:

Nilimesh Halder (nilimesh.halder@uwa.edu.au)
Joel K Kelso (joel@csse.uwa.edu.au)
George J Milne (george.milne@uwa.edu.au)

Version: 4 Date: 30 April 2014

Author's response to reviews:

As per request, we have added an acknowledge section with the following text:

This study was funded by a project grant to GM from the National Health and Medical Research Council of Australia (number 10004415). The funders had no role in study design, data collection or analysis, decision to publish, or preparation of the manuscript.